Dear Editor,

We would like to thank the Editor and Reviewers for taking the time to review our paper and for their decision to accept it for publication.

We have made all the changes requested by the first Reviewer and these appear in the revised text in bold letters and underlined. Figure 1 refers to the preoperative x-rays when the curves had progressed from initial clinical assessment. We have added Figure 1 under Case presentation, paragraph 6 in order to clarify the Reviewer's question.

Thank you again for your help.

Kind regards,

Yours sincerely,